Author's response to reviews

Title: Understanding the acceptability of e-mental health - attitudes and expectations towards computerised self-help treatments for mental health problems

Authors:

Peter Musiat (peter.musiat@kcl.ac.uk)
Philip Goldstone (philip.goldstone@kcl.ac.uk)
Nicholas Tarrier (nicholas.tarrier@kcl.ac.uk)

Version: 2 Date: 21 November 2013

Author's response to reviews:

We added acknowledgements before the references. The section reads:

This work is supported by the Department of Health via the National Institute for Health Research (NIHR) specialist Biomedical Research Centre for Mental Health award at South London and Maudsley NHS Foundation Trust and Institute of Psychiatry, King’s College London. The views expressed herein are not necessarily those of the NHS, the NIHR or Department of Health. We thank the member of the Service User Advisory Group at the Institute of Psychiatry for their contribution to the development of this study.